Author’s response to reviews

Title: Combined phacoemulsification and viscocanalostomy with Ologen implant versus combined phacoemulsification and viscocanalostomy

Authors:

Ahmed Gad (ahmedmoemen@yahoo.com)
Bahaa-Eldin Hasan Abdulhalim (bahaah2001@hotmail.com)
Ayman Lotfy (elnadyayman@gmail.com)
Ayman Mohamed Abdelrahman (aayman_1@hotmail.com)
Ahmed Samir Ahmed (ahmedsamir74@gmail.com)

Version: 2 Date: 24 Jan 2019

Author’s response to reviews:

Editor's comment 1:

Please include a section entitled "Trial Registration" after the Conclusion section of your manuscript. Please include your trial registration number (as discussed in previous emails) along with the date of registration and the statement "retrospectively registered.

Response:
Done (page 16 line 3)

Editor's comment 2:

Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Response:
Done (page 17 lines 9, 10, 15, 16, 19, 20, 22 and page 18 line 3)
Editor's comment 3:

Please remove the authorship change form, responses to reviewers and the English editing certificate as they are no longer needed at this point.

Response:

Done

Editor's comment 4:

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response:

Done